Author's response to reviews

Title: Gender differences in health care use among the elderly population in areas of Norway and Finland. A cross-sectional analysis based on the HUNT Study and the FINRISK Senior Survey.

Authors:

Anna Liisa Suominen-Taipale (liisa.suominen-taipale@ktl.fi)
Tuija Martelin (tuija.martelin@ktl.fi)
Seppo Koskinen (seppo.koskinen@ktl.fi)
Jostein Holmen (jostein.holmen@ntnu.no)
Roar Johnsen (roar.johnsen@ntnu.no)

Version: 4 Date: 1 September 2006

Author's response to reviews:

Dr Anna Liisa Suominen-Taipale
Department of Health and Functional Capacity
KTL (National Public Health Institute)
Mannerheimintie 166, FIN-00300 Helsinki,
FINLAND P.O. Box 5281
E-mail: liisa.suominen-taipale@ktl.fi

To the BioMed Central Editorial Team,

Please find followed by this cover letter our revised manuscript of a research article titled "Gender differences in health care use among the elderly population in areas of Norway and Finland. A cross-sectional analysis based on The HUNT Study and the FINRISK Senior Survey ".

Description of the changes that were made found below:

- Figure and table placement texts removed.
- A sentence 'All the authors have read and approved the final manuscript' added in authors' contributions.
- Reference list checked and notes 'et al.' removed.

Yours sincerely,

Anna Liisa Suominen-Taipale
Tuija Martelin
Seppo Koskinen
Jostein Holmen
Roar Johnsen